Response to Reviewers’ comments

Dear Editor,

We would like to thank you very much for your valuable and thoughtful comments that greatly helped to improve our manuscript. We have carefully considered your valuable comments and good suggestions. In the following, we are going to explain how your comments have been taken into full account in the revision.

Reviewer #1:

Scientific Quality: Grade C (Good)
Language Quality: Grade B (Minor language polishing)
Conclusion: Minor revision

Specific Comments to Authors: Comment #1: please add a paragraph in the method section, confirming IRB approval

Response 1 We appreciate it very much and we have carefully checked and improved the English writing in the revised manuscript. A paragraph has been added in the method section.

Finally, thank you so much again for all your valuable comments and suggestions. We hope that the revised manuscript has addressed all the concerns and is worth publishing.

Thank you very much!
Best regards,
Bo-Wen Zhao, MD.
Sir Run Run Shaw Hospital, Zhejiang University College of Medicine
Hangzhou
Zhejiang, China
Email: zbwcjp@zju.edu.cn